Author's response to reviews

Title: Fractal dimension of chromatin is an independent prognostic factor for survival in melanoma

Authors:

Valcinir Bedin (drbedin@uol.com.br)
Randall L Adam (randalladam@yahoo.com.br)
Bianca CS de Sa (bianca.sa@terra.com.br)
Gilles Landman (glandman@terra.com.br)
Konradin Metze (kmetze@pq.cnpq.br)

Version: 5 Date: 27 May 2010

Author's response to reviews: see over
Dear Sirs,

Regarding your concerns I would like to answer:

1. All authors have read and approved the final manuscript.
2. Figure legends provided
3. Vertical lines removed
4. Figure cropping done